Author's response to reviews

Title: Using the theory of planned behaviour as a process evaluation tool in randomised trials of knowledge translation strategies: A case study from UK primary care

Authors:

Craig R Ramsay (c.r.ramsay@abdn.ac.uk)
Ruth E Thomas (r.e.thomas@abdn.ac.uk)
Bernard L Croal (bernie.croal@nhs.net)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Martin P Eccles (Martin.Eccles@newcastle.ac.uk)

Version: 7 Date: 13 September 2010

Author's response to reviews:

As requested,
Title and abstract amended on online system;
References amended to include all authors;
Figures cropped to remove white space;
Figure 2 legend moved to main manuscript.
Regards
Craig